EO-1 Weekly Status Week of April 6 – April 12, 2007

Day of Year 096 - 102
Mission Day 2326 - 2332
Earth Observing-One (EO-1) General:

There were 146 Data Collection Events (DCEs) scheduled this week.  

The instruments underwent decontamination cycling as follows:

HSI performed a deicing from 04/12/2007 at 22:19z (07-102) to 04/13/2007 at 13:05z (07-103).  

ALI performed an outgassing from 04/12/2007 at 22:27z (07-102) to 04//2007 at 13:19z (07-103).  

EO-1 Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) anomaly (see below).

C&DH 

Per NASA Headquarters direction, effort has continued to develop and implement workarounds for the Command and Data Handling Medium Speed Serial Port anomaly that prevents normal playback of recorded engineering data during Ground Network (GN) contacts.  The strategy being employed allows meeting 100% of science imaging requirements and technology testbed activities while pursuing the workarounds.
Instruments:

The instruments operated nominally this week. 

Technology Activities:
In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

An AIST Sensor Web collaborators team telecon was held April 12, 2007.  During this telecon, the following principle items for the fire demo were discussed: 
· Two major steps have been taken to produce both Level 0 and Level 1G data within a few hours of being downlinked.  The Level 0 data is being produced at GSFC and then delivered to JPL where Level 1G data is produced using code developed by USGS/EROS.  These accomplishments are the first steps to making these data processing capabilities into an on-demand web service component of our web services architecture. 
· A commitment has been obtained from the JPL ASTER team to allow us to task ASTER to acquire targets within the UAS flight corridor and return expedited data within a few hours after downlinking the data to the Forest Service RSAC facility in Salt Lake, UT.
· There were discussed particular UAS fight constraints (e.g., turning radius and banking angle) and how to incorporate them into the fire demo optimizer planner.  
· The first draft of science and operations requirements, based primarily on use case descriptions already developed for the fire demo, will be delivered next week 
· A new sensor web use case was initiated for the emergency detection of Rhodamine dye in water that would allow support of maritime search and rescue missions. 
· A letter has been prepared and forwarded to the Air Force requesting official use of Draper Lab produced cloud predicts for use in the fire demo.
· All the interfaces required for an EO-1 workflow that supports the fire demo have been identified.  Some of these have been fully specified while the remaining ones are currently under development. 
Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 all week.

Ground System

Hardware Status

	Backup R/T System
	
	Yellow
	A new PC will be ordered to create a backup string file server for real-time operations.  Request and a quote for this new system were sent to the project on 3/8/2007.  Also, Red-Hat licenses are being obtained for ASIST workstations.  

	Mission Planning
	
	Yellow
	Oracle issues on speedwheel have been resolved.  FOT currently is performing most mission planning tasks on the backup machine Speedwheel.  Speedwheel’s configuration will be duplicated to build the primary machine Axle.  FOT is currently testing all mission planning tasks on Speedwheel to ensure that expected results are obtained.  VMWare licenses are being obtained for all the mission planning machines.

	Flight Dynamics
	
	Green
	FOT is waiting for the arrival of two new PCs to the MOCC that will be used for upgrading flight dynamics software and tasks.  These new PCs will be configured to move away from HP UNIX.


Operations

EO-1 FOT has been meeting with the systems administration group to address issues that currently effect operations.  In addition to the actions identified above for Hardware Status, data backup methods and failover system capability will be set up and tested independently for the mission planning, real-time, and flight dynamics machines.     
Ground and Space Network


EOS Data and Operations System (EDOS), GSFC,MD,USA

There are no major problems to report.

McMurdo Ground Station (MGS), McMurdo, Antarctica




There are no major problems to report.

Wallops Ground Station (WGS), Wallops Island, VA, USA


There are no major problems to report.

DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.

Alaska Ground Station (AGS), Poker Flat, AK, USA 

There are no major problems to report.


Svalbard Ground Station (SGS), Longyearbyen, Norway
There are no major problems to report this week.

Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.

USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA
 LGS is now being scheduled only for contingencies.

Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

Australian Centre for Remote Sensing reports that TERSS has been having some technical issues.  The antenna will be down for the next two weeks.

Cordoba Ground Station (CGS), Cordoba, Argentina


No passes were scheduled this week. 

Upcoming Events:

· Orbit Lowering maneuver to occur on May 1, 2007 at 13:18:03 GMT.

Imagery Status:

Scenes and Engineering Cals planned for week of April 6 – April 12, 2007                    146
Total scenes and engineering calibrations planned for entire mission – approximately 33,645
Total Scenes:  ALI scenes in the Level 0 archive            30,549 (as of April 12, 2007)

                        Hyperion scenes in the Level 0 archive    30,297
Publications and Presentations Status (as of 03-30-07):

341 publications 

240 external presentations

54 articles and press releases


































































































































































































































































































































































































































































































































































































Page 2 of 3

